Author's response to reviews

Title: The development of a multidisciplinary fall risk evaluation tool for demented nursing home patients in the Netherlands

Authors:

Jacques C.L. Neyens (j.neyens@irv.nl)
Beatrice P.J. Dijcks (b.dijcks@irv.nl)
Jolanda C.M. van Haastregt (j.vanhaastregt@zw.unimaas.nl)
Luc P. de Witte (lp.dewitte@irv.nl)
Wim J.A. van den Heuvel (w.vandenheuvel@irv.nl)
Harry F.J.M. Crebolder (harry.crebolder@hag.unimaas.nl)
Jos M.G.A. Schols (jschols@vivregroep.nl)

Version: 4 Date: 23 December 2005

Author's response to reviews:

Dear Editor,

Thank you for (provisionally) accepting our paper for publication in BMC Public Health.

According your suggestions we made the following corrections to our text:
- we revised the title page and text according to the manuscript formatting checklist
- in the abstract we removed the coma after developed in the first sentence of the Results and we changed the last subheading from Conclusion to Conclusions.
- we removed the shading from the tables
- based on the suggestion of Reviewer Ruchinskas we added some sentences about the interventions which could follow the fall risk evaluation, at the end of the Results section.

I hope our revisions will make publication of our paper possible.

Best wishes,

Jolanda van Haastregt